

# Water Resources Research<sup>®</sup>

## **RESEARCH ARTICLE**

10.1029/2023WR035590

#### **Key Points:**

- Downstream nitrate & ammonium concentrations were seasonally responsive to flow, while total phosphorus & orthophosphate were chemostatic
- Nutrient concentrations were strongly related to watershed inputs, and less consistently to regulatory releases
- Restoring downstream water quality will require reservoir and watershed management, and actions that go beyond modified release scheduling

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

N. G. Nelson and L. R. Montefiore, nnelson4@ncsu.edu; lrmontefiore@ncsu.edu

#### Citation:

Montefiore, L. R., Kaplan, D., Phlips, E. J., Milbrandt, E. C., Arias, M. E., Morrison, E., & Nelson, N. G. (2024). Downstream nutrient concentrations depend on watershed inputs more than reservoir releases in a highly engineered watershed. *Water Resources Research*, 60, e2023WR035590. https://doi.org/10.1029/ 2023WR035590

Received 21 JUN 2023 Accepted 27 FEB 2024

#### **Author Contributions:**

Conceptualization: L. R. Montefiore, D. Kaplan, E. J. Phlips, E. C. Milbrandt, M. E. Arias, E. Morrison, N. G. Nelson Data curation: L. R. Montefiore Formal analysis: L. R. Montefiore Funding acquisition: D. Kaplan Methodology: L. R. Montefiore, D. Kaplan, M. E. Arias, N. G. Nelson Project administration: D. Kaplan Software: L. R. Montefiore Supervision: N. G. Nelson Visualization: L. R. Montefiore

© 2024. The Authors. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# Downstream Nutrient Concentrations Depend on Watershed Inputs More Than Reservoir Releases in a Highly Engineered Watershed

L. R. Montefiore<sup>1</sup>, D. Kaplan<sup>2</sup>, E. J. Phlips<sup>3</sup>, E. C. Milbrandt<sup>4</sup>, M. E. Arias<sup>5</sup>, E. Morrison<sup>2</sup>, and N. G. Nelson<sup>1,6</sup>

<sup>1</sup>Biological and Agricultural Engineering, North Carolina State University, Raleigh, NC, USA, <sup>2</sup>Environmental Engineering Sciences, University of Florida, Gainesville, FL, USA, <sup>3</sup>Fisheries and Aquatic Sciences, University of Florida, Gainesville, FL, USA, <sup>4</sup>Marine Laboratory, Sanibel-Captiva Conservation Foundation, Sanibel, FL, USA, <sup>5</sup>Department of Civil & Environmental Engineering, University of South Florida, Tampa, FL, USA, <sup>6</sup>Center for Geospatial Analytics, North Carolina State University, Raleigh, NC, USA

**Abstract** In this study, we characterized the impact of regulatory water releases relative to watershed inputs on the quality of receiving waters to identify if and how managed releases could be scheduled to mitigate nutrient export and downstream water quality impairment. We specifically investigated freshwater flow partitioning to the Caloosahatchee River and Estuary (CRE) from a large managed lake, Lake Okeechobee, and the CRE's upstream watershed, the C-43 basin, in southwest Florida (USA). A water balance was developed to identify dominant freshwater inflow sources (i.e., Lake Okeechobee vs. watershed inputs) over time. From the water balance, analyses of historical trends were performed to detect changes in freshwater inflow contributions to the CRE. Further, seasonal and annual concentration variations and longterm concentration-discharge (C-Q) relationships were analyzed to better understand biogeochemical and hydrological processes in the system in relation to freshwater source. Since 1966, we found the duration and magnitude of flows from the C-43 basin were higher than those from Lake Okeechobee releases. However, recent increases in the annual water volume and proportion of inflow coming from Lake Okeechobee to the CRE were observed. The C-Q analysis revealed that nitrate and ammonium concentrations in the CRE were responsive to changes in discharge, while total phosphorus and orthophosphate concentrations were chemostatic. While modifications to the Lake Okeechobee operation schedule could potentially mitigate downstream inorganic nitrogen loading, this potential is limited by complex, seasonal C-O relationships and confounding effects from surrounding watersheds.

**Plain Language Summary** Water levels in large lakes at risk of flooding into surrounding areas are controlled using engineered structures like dams. To ensure water levels do not overtop a lake's banks, water is released from control structures and then flows into downstream waterways like rivers and estuaries. When a managed lake has water quality challenges, such as excess nutrient concentrations or harmful algal blooms, water releases may affect the quality of downstream waters, but determining the role of released waters on downstream water quality is challenging. In this study, we analyzed nitrogen and phosphorus concentrations in the Caloosahatchee River and Estuary (CRE), a waterway that receives released water from Lake Okeechobee, a large managed lake, to understand whether there were relationships between Lake Okeechobee water releases and nutrient concentrations in the CRE. We found that released water had an impact on downstream water quality, but that water runoff from the surrounding land area had a greater effect. Nitrogen concentrations varied based on the time of year and amount of flowing water, while phosphorus concentrations did not. Therefore, changes to the timing and volume of water releases may not affect downstream phosphorus concentrations, but could potentially improve downstream nitrogen concentrations.

### 1. Introduction

Human settlement has profoundly altered the natural landscape, increasing environmental pressures on watersheds across the globe. At the same time, demand for freshwater, energy and other resources has increased, driven by the growing needs of diverse sectors, including municipalities, agriculture, and industry (Dahm, 2010; Holland et al., 2015; Vörösmarty et al., 2010). To meet these needs, thousands of reservoirs have been constructed, often by damming and diking natural lakes and rivers (Chen et al., 2016; WCD, 2000).



Writing – original draft: L. R. Montefiore Writing – review & editing: L. R. Montefiore, D. Kaplan, E. J. Phlips, E. C. Milbrandt, M. E. Arias, E. Morrison, N. G. Nelson Today, there are more than 90,000 dams in the United States (USACE, 2022) and more than 500,000 large dams in the world (Downing et al., 2006).

Management of reservoirs and lakes has greatly modified the quality and quantity of freshwater inflow to downstream waterbodies (Benson, 1981; Cerco & Noel, 2016; Ling et al., 2017; Nilsson et al., 2005), which affect their structure and function (Grill et al., 2019; Lehner et al., 2011; Nilsson et al., 2005; Poff et al., 2007; Wu et al., 2019; Yi et al., 2010). Water quality degradation can be especially pronounced in downstream water bodies receiving large and frequent discharges containing high levels of nutrients and/or pollutants (Liu et al., 2009). Lakes and rivers located in human-dominated landscapes are particularly susceptible to loading from a wide range of nutrients and pollutants from agricultural, industrial, and urban land uses (Huang et al., 2013; Klatt et al., 2003; Motew et al., 2017; Rechcigl, 1997). Pollutant loads can accumulate in these systems, leading to discharges to downstream waters that result in significant water quality impacts, such as harmful algal blooms (Glibert et al., 2009; Paerl et al., 2008; Phlips et al., 2012, 2023). The challenge of managing water quality in water bodies subject to large discharges from developed watersheds is further complicated by the presence of dams, dikes and other water control structures, which alter hydrologic conditions. One of these challenges is the need to determine the extent to which downstream nutrient and pollutant loads are sourced directly from engineered water control structures versus more diffuse watershed inputs. Fortunately, multi-year time series of water quality and hydrology are now available for many aquatic systems across the US and can offer insights on how downstream water quality responds to different flow regimes and sources of water.

Long-term observations can be used to characterize concentration-discharge (C-Q) relationships, which describe a watershed's hydrological behavior and dominant biogeochemical processes (Godsey et al., 2009; Musolff et al., 2021). Depending on the slope of the C-Q relationship, three solute export patterns can be characterized: dilution, mobilization, and chemostasis (Godsey et al., 2009). Several studies have investigated C-Q relationships in forested and agricultural watersheds (Duncan et al., 2017; Godsey et al., 2009; Liu et al., 2022). However, research on C-Q relationships in highly engineered and managed watersheds has been comparatively limited, though relationships between nutrient loads and flows in some human-dominated basins have been investigated (e.g., Basu et al., 2010). Studying C-Q relationships in waters receiving regulatory releases from reservoirs or managed lakes could help to define the extent to which releases are responsible for worsening downstream water quality, as compared to the impacts of diffuse watershed inputs, and potentially inform operation schedules. In particular, the application of C-Q relationships to ecosystems with water control structures could be used to optimize operation schedules such that they improve downstream water quality (or degrade it less). For example, if time periods when nutrient mobilization is likely to occur through water releases were identified, operation schedules could be adjusted to avoid major discharges during these times, and instead prioritize releases when dilution is more likely to occur.

To determine the impact of regulatory water releases relative to watershed inputs on receiving waters and identify how releases could be scheduled to mitigate nutrient export and downstream water quality impairment, we analyzed trends in controlled releases from a large, managed, and eutrophic lake in Florida (Lake Okeechobee) and its downstream waterway (Caloosahatchee River and Estuary, CRE), located in southwest Florida. Our objectives were to: (a) Evaluate how operations schedules impacted downstream waters, (b) quantify the relative contribution of regulatory water releases on total flows in the downstream receiving waterbody, and (c) estimate how nutrient concentrations in receiving waters vary as a function of freshwater source (i.e., regulatory lake releases vs. diffuse watershed inputs to the CRE) across years and seasons. In addition to producing findings that can inform how Lake Okeechobee's operation schedule could be designed to mitigate downstream water quality impacts, the presented approach offers generalizable findings as to how analyses of observational data can provide practical insight for defining environmental flow criteria and lake-reservoir operational schedules.

## 2. Materials and Methods

### 2.1. Study Area

Lake Okeechobee is a large (1,890 km<sup>2</sup>), shallow (mean depth 2.7 m) lake in South Florida. It is a highly eutrophic lake with substantial internal nutrient loads and recurring intense harmful algal blooms, including of the cyanobacteria *Microcystis aeruginosa*, *Dolichospermum* spp. (formerly *Anabaena*), *Raphidiopsis raciborskii* (formerly *Cylindrospermopsis raciborskii*) (Kramer et al., 2018; Phlips et al., 2020). Although of natural origin, the lake is surrounded by a human-made dike and operated as a reservoir, making it central to the hydrology and





Figure 1. Study area illustrating the Caloosahatchee River and Estuary (CRE) and the S-77, S-78, and S-79 lock and dams. The CRE watershed is composed of the C-43 basin (upstream of S-79) and the tidal basin (downstream of S-79).

water management of the region (SFWMD, 2009). Historically, Lake Okeechobee was the primary source of freshwater to the Everglades, an extensive sawgrass marsh complex that stretched across most of the southern peninsula of Florida. To prevent deadly flooding, as well as facilitate navigation and agriculture, the U.S. Army Corps of Engineers (USACE) constructed over 30 water control structures since the 1930s to manage Lake Okeechobee water levels and a canal system to divert Lake Okeechobee discharges to the east and west coasts of Florida. Over the past 30 years, an average of 38% (±13% SD; range: 11%–59%) of total Lake Okeechobee annual outflows have been artificially discharged to the Caloosahatchee River Estuary (CRE) on Florida's southwest coast (Figure 1); average flows to the St. Lucie and south to the Everglades account for 17% and 45% of Lake Okeechobee discharges, respectively (Zacharias & Kaplan, 2023). While successful in preventing catastrophic floods, controlled water releases from Lake Okeechobee fundamentally transformed the hydrology and ecology of South Florida. Today's Everglades span only 50% of the original area (Sklar et al., 2005), resulting in the loss of significant amounts of water storage on the southern Florida peninsula.

Several Lake Okeechobee operation schedules have been implemented over the past few decades, with the schedules dictating when and how much water is discharged to the CRE and other downstream ecosystems. Since 1978, five operation schedules have been authorized to control floods and meet other demands, with each successive plan increasing in complexity in response to growing concerns over an expanding human population and declining ecosystem health (Table 1) (Julian & Reidenbach, 2023; Tarabih & Arias, 2021). Currently, under the Lake Okeechobee Regulation Schedule (LORS), Lake Okeechobee releases are "pulsed" to the CRE in the dry season on a regular basis. This approach was adopted to mitigate the occurrence of hypersaline conditions in the CRE (Tarabih & Arias, 2021), but it may have the unintended consequence of providing persistent nutrient loads to the estuary.

Three lock and dam structures, referred to as the S-77, S-78, and S-79 structures, are used to release water from Lake Okeechobee to the CRE and maintain water levels in the C-43 canal (Figure 1). Lock S-77 serves as an outlet from Lake Okeechobee, and S-79 is 68 km downstream of S-77 and serves as a salinity and tide barrier at the upstream boundary of the CRE (SFWMD, 2009). The C-43 canal connects S-77–S-79, and receives watershed inputs from the surrounding basin. The S-78 structure was constructed for navigation, irrigation, flood, drought, and regulatory control, and is located between S-77 and S-79 on the C-43 canal. In addition to direct inputs from Lake Okeechobee, the CRE receives surface runoff from a 4,370 km<sup>2</sup> watershed (SFWMD, 2009) that includes the C-43 basin (70% of watershed land area) and the "tidal" basin (30% of watershed land area) (Julian & Osborne, 2018). Lake Okeechobee releases and C-43 watershed inputs both arrive to the CRE via the S-79

#### Table 1

Authorized Operations Schedules of Lake Okeechobee Since 1978 (Cadavid et al., 2012; Tarabih & Arias, 2021)

Lake Okeechobee operation schedule	Date of implementation	Objectives
1978 Rules	1978	• Maintain water levels between 4.72 and 5.64 m
Run 25	1991	• Maintain water levels between 4.77 and 5.11 m
Water Supply and Environment (WSE)	2000	• Maintain water levels from 4.1 to 5.6 m
		• Decrease water discharges into the estuaries
		• Decrease the frequency of littoral zone flooding
		• Meet water user's demands, while allowing water levels to drop lower than antecedent schedule
Lake Okeechobee Regulation Schedule (LORS) 2008	2008	• Prevent lake level exceedance above 4.88 m
		• Identify seasonal high-water levels for which water is discharged to prevent saltwater intrusion during the dry season
System Operating Manual (LOSOM)	Plan under final stages of review	• Incorporate flexibility in water management to better balance different needs (e.g., navigational, flood control, preservation of fish and wildlife)

structure, and the tidal basin drains to the CRE downstream of S-79 through a number of small creeks (Figure 1). The primary land use in the C-43 watershed is irrigated agriculture, accounting for 27% of the area, followed by pasture and hay at 23%, and urban land use at 6%, according to the 2019 National Land Cover Database (NLCD) (Dewitz, 2021). In contrast, residential and commercial developments are the predominant land uses in the tidal basin (SFWMD, 2009). Analyses carried out by the South Florida Water Management District (SFWMD) revealed that nutrient loads from the C-43 basin were much higher than those coming from the tidal basin (SFWMD, 2009). However, additional research is needed to separate the effects of C-43 watershed inputs and Lake Okeechobee releases on water quality in the region to better understand the impact of regulatory water releases relative to watershed inputs.

### 2.2. Flow and Water Chemistry Data

Average daily discharge from the S-77 and S-79 structures were obtained from DBHYDRO, a database maintained by the SFWMD (accessed at: https://my.sfwmd.gov/dbhydroplsql/show\_dbkey\_info.main\_menu). The data set includes daily mean flow observations dating back to 1966, and the period of record analyzed here spanned from January 1966 to August 2022. Daily rainfall observed at S-79 was also retrieved from DBHYDRO from 1965 to 2022. The S-78 data record was not considered in the analysis due to limited records compared to S-79.

Water quality parameters were also accessed via DBHYDRO, specifically for orthophosphate (Ortho-P), total phosphorus (TP), total nitrogen (TN), dissolved phosphorus (DP), nitrate and nitrite (NO<sub>x</sub>), and ammonium  $(NH_4^+)$  concentrations. Flagged observations that did not meet quality control checks were excluded. For consistency, only surface samples (depth ≤0.5 ft) were considered. Nutrient concentrations were compared against reported Minimum Detection Limits (MDL); when an observation was less than its MDL, the value was replaced with the MDL concentration, and an observation was excluded if no MDL was reported. 2% of  $NH_4^+$  (n = 13), 15% of NO<sub>x</sub> (n = 92), 1% of Ortho-P (n = 6), and 1% of TP (n = 6) samples were below the MDL. If multiple sample values were reported for the same analyte on the same day, these values were averaged. After screening and cleaning, water quality data were available from 803 sampling events at the S-79 structure. Prior to May 2010, most samples were collected monthly. However, starting in May 2010, most samples were collected approximately weekly, and in 2016, the sampling frequency changed to approximately biweekly. Accordingly, nutrient concentrations prior to May 2010 were excluded from subsequent analysis as data were too scarce. Of the available sampling dates from May 2010 to August 2022 (n = 605), NH<sub>4</sub><sup>+</sup>, NO<sub>8</sub>, Ortho-P, and TP had few missing observations (n = 20, 14, 3, and 1, respectively), whereas TN and DP had many more missing values (n = 215 and 437, respectively) and were thus excluded from analysis. The final data set included 585 observations of  $NH_4^+$ , 591 of NO<sub>x</sub>, 602 of Ortho-P, and 604 of TP.

### 2.3. Water Balance

A daily water balance was developed to identify the dominant source of freshwater inflow to the CRE (i.e., C-43 basin runoff vs. Lake Okeechobee releases). Flow from the C-43 basin was assumed to come via the C-43 canal and was calculated as the difference between S-79 and S-77 discharge ( $Q_{C-43} = Q_{S-79} - Q_{S-77}$ ) for periods when the following criteria were met: (a) Flow at S-77 was positive (i.e., no backflow into the lake); and (b) the difference between S-79 and S-77 flows was positive (i.e., flow was greater at S-79 than S-77). Backflow from S-77 into Lake Okeechobee only occurred for 3% of the period of record. As a simple metric of flow dominance, we calculated the flow ratio between C-43 inputs and total flow at S-79. When this value was greater than or equal to 0.5, the dominant source of flow was from the C-43 watershed; when this ratio was less than 0.5, the dominant flow source to the estuary was from Lake Okeechobee releases. Discharge dominance was analyzed over the overlapping period of record for the S-77 and S-79 stream gauges stations and broken out over three hydroclimatological seasons defined by Julian and Osborne (2018): Dry = November through May; Early Wet = May through July; and Wet = August through October.

### 2.4. Trend and Concentration-Discharge (C-Q) Analyses

To investigate the hydrological and biogeochemical characteristics of the watersheds, we performed trend and C-Q analyses. To gain a comprehensive understanding of the role of freshwater source on nutrient concentrations in the CRE, seasonal variation within and across years needed to be considered to account for intra- and inter-annual hydrologic dynamics. Therefore, we assessed nutrient concentrations in relation to freshwater source within seasons and years.

The trend analysis was conducted using the rank-based, non-parametric Mann-Kendall test (MK) (Kendall, 1948; Mann, 1945) using the statistical software R (R Core Team, 2020) to test for temporal trends in annual average daily discharge proportion and the annual average daily volume attributed to Lake Okeechobee at the S-79 stream gauge between 1966 and 2022. The non-parametric MK method is less sensitive to outliers than parametric tests and does not require specification as to whether the trend is linear or nonlinear. The Kendall Tau statistic, which varies from -1 to 1, measures the monotonicity of the slope, with +1 being a consistently increasing trend, and -1 being a consistently decreasing trend.

Additionally, we conducted a C-Q analysis to assess the relationship between the concentrations of TP, Ortho-P,  $NH_4^+$ , and  $NO_x$  and the discharge rate at S-79 throughout the entire study period, as well as on a monthly basis. All C and Q data were log-transformed, and the linear regression slope of the log-log relationship was computed. The slope ( $\beta$ ) of the log(C)-log(Q) regression was considered significantly different from zero at  $\alpha = 0.05$ . A slope ranging from -0.1 to 0.1 is generally considered to correspond to chemostatic export behavior, while slopes >0.1 correspond to mobilization and those <-0.1 correspond to dilution (Godsey et al., 2009; Herndon et al., 2015).

### 3. Results

#### 3.1. Partitioning Flows to the Caloosahatchee River Estuary

Daily discharge into the CRE at the S-79 structure was highly variable over the 57-year period, with large spikes in discharge corresponding to wet years and hurricanes (e.g., Hurricane Irma in 2017) and several drought years characterized by extremely low flows (less than 0.30 m<sup>3</sup>/s). The daily discharge median was 25.4 m<sup>3</sup>/s (10th percentile = 0.28, 90th percentile =  $161 \text{ m}^3/\text{s}$ ).

Water releases from Lake Okeechobee were the dominant surface water source to the CRE for 49% of days in the period of record. Freshwater inputs from the lake were generally dominant from October through May, while runoff from the C-43 basin was usually dominant from June through September (Figure 2). However, these seasonal observations were not consistent throughout the period of record. Monthly (Figure 2b) and seasonal (Figure 2c) long-term flow observations revealed that, from the 1970s until the 2000s, annual discharge to the CRE came mainly from the C-43 basin; prior to 2003, Lake Okeechobee was the dominant source of flow to S-79 for a median of 3 months per year, increasing to 7 months per year from 2003 onwards (Figure 3a).

We found a significant increasing trend in the proportion of annual average daily discharge to the CRE coming from Lake Okeechobee (tau = 0.233, p = 0.01) as well as in the annual volume attributable to Lake Okeechobee at the S-79 station (tau = 0.206, p = 0.02) (Figures 3b and 3c). The changes in source partitioning appeared to







**Figure 2.** S-79 freshwater discharge heatmap for the Julian day (a), month (b), and season (c), and annual precipitation at S-79 (d). The color ramp in panels (a)–(c) corresponds to the proportion of S-79 discharge attributed to watershed inputs relative to Lake Okeechobee (Lake O) water releases. For values greater than 0.5 (i.e., more yellow), the dominant source of flow comes from the watershed (C-43 basin) and for values less than 0.5 (i.e., more blue), the dominant source of flow comes from Lake Okeechobee. From 1987 to 1992, over 30% of the daily precipitation data were missing, therefore, annual rainfall was reported as NA for these years (d).

coincide with the Lake Okeechobee regulatory management change from Run25 to WSE (Table 1), the adoption of which resulted in regulatory releases occurring more frequently (Figure 2). However, S-79 hydrographs from individual years (Figure 4) show that the duration and magnitude of flows from the C-43 basin were generally higher than those from Lake Okeechobee releases. While Lake Okeechobee was more frequently the dominant flow source on a daily basis (i.e., Lake Okeechobee was the dominant freshwater source to the CRE on most days in the period of record), the C-43 basin was the primary source of freshwater inflow to the CRE *by volume* for 41 of 57 years in the period of record (Figure 3c). Of note, although C-43 watershed inputs dominated the total freshwater flow volumes to the CRE, most freshwater flow from C-43 occurred during the wet season (Figure 2c).

#### 3.2. Nutrient Responses in Relation to Dominant Water Source

For TP and Ortho-P, similar monthly patterns were observed, with the lowest concentrations observed from November to May (dry season), which corresponded to periods when Lake Okeechobee was the dominant source of inflow. Higher concentrations were observed for the rest of the year (Figure 5), when the C-43 basin was the primary source of freshwater inflow. For NO<sub>x</sub>, higher concentrations were observed in November (median of 0.32 mg/L) and the lowest concentrations were observed in May (median of 0.018 mg/L). For NH<sub>4</sub><sup>+</sup>, the concentration was consistent over the year (Figure 5).

When considering all data, statistically significant patterns of mobilization were observed for all water quality parameters, except TP, for which chemostasis was observed ( $\beta_{TP} = 0.09$ ). The strongest effects were observed for NH<sub>4</sub><sup>+</sup> ( $\beta_{NH_4^+} = 0.37$ ) and NO<sub>x</sub> ( $\beta_{NO_x} = 0.25$ ), while the effects were relatively weak for Ortho-P ( $\beta_{Ortho-P} = 0.11$ , i.e., just above the threshold for chemostasis) (Figure 6). These results indicate there was little influence of discharge on variation in TP and Ortho-P concentrations but a stronger influence on NH<sub>4</sub><sup>+</sup> and NO<sub>x</sub> concentrations. In particular, NH<sub>4</sub><sup>+</sup> exhibited mobilization behavior for all the months of the year except from March to May, for which insignificant relationships were found (Figure 7). The dominant source of inflow does not seem to affect the mobilization pattern of NH<sub>4</sub><sup>+</sup> (Figure 6). Contrary to NH<sub>4</sub><sup>+</sup>, NO<sub>x</sub> showed both mobilization and dilution behaviors; significant mobilization patterns were observed in February and April to





**Figure 3.** Number of months where Lake Okeechobee (Lake O) was the dominant source at S-79 (a), average daily discharge proportion (%) attributable to Lake Okeechobee (MK test: tau = 0.233, *p*-value = 0.01) and C-43 watershed at S-79 (b), and average daily volume attributable to Lake Okeechobee (MK test: tau = 0.206, *p*-value = 0.02) and C-43 watershed at S-79 (c). All values shown on an annual basis. The vertical dashed lines correspond to different Lake Okeechobee operation schedules (Table 1). The horizontal red dotted line at 50% (panel b) is used to differentiate dominant flow source.

July, while significant dilution patterns were observed from September to November (Figure 7). The dominant source of freshwater inflow to the CRE was C-43 basin runoff from June to October and Lake Okeechobee for the rest of the year (Figure 2b). Thus, temporal patterns observed in  $NO_x$  mobilization and dilution do not perfectly align with periods of either dominant C-43 basin runoff or Lake Okeechobee releases.

### 4. Discussion

Designing optimal operating strategies for reservoirs and managed lakes to meet water supply demand and flood prevention needs without impacting downstream water quality is challenging. This study aimed to better understand the impacts of regulatory water releases of a managed lake (Lake Okeechobee) on a downstream estuarine system (the Caloosahatchee River and Estuary; CRE) to identify if releases could be scheduled to mitigate nutrient export and downstream water quality impairment. We quantified the relative contribution of the different freshwater inflow sources (i.e., regulatory water release from the lake vs. surface runoff from the surrounding watershed), and evaluated how nutrient concentrations varied as a function of freshwater source.





Jan Feb Mar Apr May Jun Jul Aug Sep Oct Nov Dec Jan

**Figure 4.** S-79 freshwater discharge for the years 1990 (a) and 2006 (b). Points are colored by the dominant source of freshwater inflow, with yellow corresponding to C-43 basin inputs and blue to Lake Okeechobee (Lake O) water releases. Relatively rainfall-driven discharge patterns were observed for some years (e.g., 1990 corresponding to the 1978 Rules operation schedule) while others had pulsed released flows (e.g., in the dry and early wet seasons of 2006, which corresponded to the WRE operation schedule). The years 1990 and 2006 were selected as illustrative examples of system hydrology across different operation schedules.

While loads are also important to consider, we focused on concentrations given their importance for explaining biological processes such as algal growth (NRC, 2000).

Our study found that C-43 inputs are more strongly related to water quality and quantity in the CRE, particularly in the wet season, than Lake Okeechobee water releases. Prior studies arrived at the same conclusion (Caloosahatchee Estuary Basin Technical Stakeholders, 2012; Julian & Osborne, 2018). Notably, Rumbold and Doering (2020) observed that the majority of the water quality parameters (e.g.,  $NO_x$ , TP, Secchi disk depth) they studied along the CRE increased significantly in one or more regions of the estuary when the C-43 basin was the dominant source of freshwater. Our results build on the work of others by quantifying how the C-43 basin impacts water quality and quantity in the CRE relative to Lake Okeechobee using C-Q relationships, and providing more specificity on biogeochemical behaviors by characterizing intra-annual variation in mobilization, dilution, and chemostatic behaviors.



Figure 5. Monthly boxplots of NH<sub>4</sub><sup>+</sup>, NO<sub>x</sub>, orthophosphate (Ortho-P), and total phosphorus (TP) concentrations at S-79.

The C-43 basin has undergone several modifications over time to facilitate drainage and irrigation in the surrounding agriculturally dominant landscape. Rainfall-driven watershed runoff is a significant contributor to freshwater inflow in the estuary (Julian & Osborne, 2018), notably during the wet season (Figure 2). The combination of artificial drainage and intensive agricultural land use likely explains why C-43 watershed inputs are more strongly related to elevated nutrient concentrations in the CRE. As far as the lesser importance of Lake Okeechobee releases on CRE nutrient concentrations, previous research proposed two potential explanations: (a) nutrients are taken up by plants in extensive marshes located between S-77 and Lake Okeechobee and (b) water residence times are long in Lake Okeechobee, which would allow biological and chemico-physical processes to remove nutrients from the water prior to being discharged to the CRE. Although these mechanisms may help to explain why Lake Okeechobee releases are less consequential in driving elevated nutrient concentrations in the CRE, managed releases still play a major role, and Lake Okeechobee remains a considerable source of nutrients that must be considered in management actions.

Although the C-43 basin was the major contributor of freshwater volume to the CRE over the period of record, Lake Okeechobee releases are highly consequential as a driver of CRE hydrology (Figure 2). Lake Okeechobee operation schedules have evolved over the last decades to improve downstream conditions, prevent flooding, and increase the water supply. These operational changes (Table 1) were associated with an increasing trend in Lake





**Figure 6.** Nutrient concentrations (*y*-axis) and daily discharge at S-79 (*x*-axis) plots using the log-scale. The dominant source of flow (blue for Lake Okeechobee (Lake O, yellow for watershed) was determined through the water balance. Significant mobilization patterns were observed for  $NH_4^+$  and  $NO_x$  and chemostatic behaviors were observed for orthophosphate (Ortho-P) and total phosphorus (TP). The slopes ( $\beta$ ) of the log(C)–log(Q) regressions were all significant (*p*-value < 0.05).

Okeechobee water releases discharged to CRE, specifically starting in the late 1990s and early 2000s (Table 1, Figure 2). This increase aligns with prior studies demonstrating how the LORS2008 schedule was responsible for most of the increasing trends in water flows to the CRE (Tarabih & Arias, 2021). Prior to the implementation of LORS2008, flow releases from Lake Okeechobee were more variable, after which they became more frequent and regularly pulsed (e.g., Figure 4). LORS2008 was adopted to mitigate impacts to the downstream estuary by decreasing the maximum lake stage and releasing smaller volumes of water more frequently, as opposed to large volumes of water all at once. These smaller water releases aimed to create a salinity gradient in the estuary that would support sea grasses and oyster reefs, which are particularly vulnerable to periods of prolonged elevated salinity (Doering et al., 2002; Douglass et al., 2020; Volety et al., 2009). Thus, the approach of releasing relatively small volumes of water can provide water quality benefits by mitigating deleterious increases in salinity. However, the LORS2008 modification resulted in greater total water volumes of Lake Okeechobee water being delivered to the CRE.

The recent increasing influence of Lake Okeechobee water releases through more frequent pulse releases during the dry season was also shown to correspond to changes in downstream nutrient responses, but not consistently. Our results revealed that the relationship between nutrient concentrations and discharge in the study area was complex and varied by nutrient type and season. While concentrations of Ortho-P and TP were invariant to discharge,  $NH_4^+$  showed a strong, significant, and positive response to increasing discharge in January, February, and July through November (Figure 7). In contrast,  $NO_x$  exhibited both dilution and mobilization behaviors in response to discharge depending on the season. Specifically, during the wet season (September–November), when flows were generally highest and primarily sourced from the C-43 watershed,  $NO_x$  showed dilution behavior. On the other hand, during the late dry season (April) and the transition from the early wet to wet season (July and August), when flows were lower and primarily sourced from Lake Okeechobee water releases,  $NO_x$ showed mobilization behavior (Figure 7). Observing both dilution and mobilization patterns of  $NO_x$  in the same system is in line with findings from prior research, which have reported the coexistence of these patterns in various watershed settings (Knapp et al., 2020; Moatar et al., 2017). For example, Knapp et al. (2020) observed dilution behavior during wetter conditions and mobilization behavior during drier conditions.



10.1029/2023WR035590



Figure 7. Slope (left column) and coefficient of determination ( $R^2$ ; right column) from the log(C)–log(Q) regression models for each nutrient species across months and seasons. Bars are filled by the season they correspond to; bars shown in gray with hatch marks were not significantly different from zero.

The complex and dynamic relationships observed in  $NO_x$  transport and delivery can be attributed to various driving factors, including land-use type (e.g., agriculture, urban) (Fazekas et al., 2020), hydrological characteristics (e.g., water flows, rainfall) (Anderson et al., 1997; Jones et al., 2018), and biological processes (e.g., nutrient removal or transformation by aquatic organisms) (Birgand et al., 2007). In the CRE, the observed NO<sub>x</sub> mobilization pattern may be largely attributable to agricultural operations. C-43 watershed inputs are largely sourced from agricultural lands (SFWMD, 2009), and could be delivering nutrients generated through fertilizer or manure application, explaining why high NO<sub>x</sub> concentrations were observed during the months dominated by C-43 inputs. Several studies have found positive relationships between  $NO_3^-$  and the proportion of arable lands in a watershed (Minaudo et al., 2019; Moatar et al., 2017). Due to the lack of publicly available data on nutrient application, agricultural practices were not directly studied here, but greater understanding of the timing and type of nutrient application in the C-43 basin would facilitate the development of specific and targeted strategies for mitigating nutrient export to the downstream estuary. Furthermore, water residence time can also partially explain the observed mobilization pattern of NO<sub>x</sub>. Lower flows can result in longer water residence time and increased contact time with the streambed surface, favoring  $NO_3^-$  uptake by denitrifying microorganisms (Birgand et al., 2007; Peterson et al., 2001), which could result in lower concentrations during low flow conditions. In contrast, when flow increases, especially from the C-43 basin, residence time and the potential for denitrification decrease, likely resulting in higher concentrations.

On the other hand, groundwater input may also partially explain the  $NO_x$  dilution patterns observed in the CRE in September through November. The hydraulic conductivity of the unconfined aquifer in Lee County, where the CRE is located, is high (Scott & Missimer, 2001), and submarine groundwater inputs to the CRE are substantial (Charette et al., 2013). Hence, groundwater might account for the observed dilution pattern by contributing water with lower nitrate concentrations than surface water during baseflow conditions (Rose et al., 2018). Further, riparian and biogeochemical processes occurring along the western edge of Lake Okeechobee, which is composed of a large wetland marsh area (Figure 1), can filter water before it leaves the lake and therefore might result in lower  $NO_x$  concentrations in Lake Okeechobee releases (Doering & Chamberlain, 1999). Previous research has posited that biogeochemical processes, including nitrate uptake and denitrification within streams or riparian zones, exert a more substantial relative impact on nitrate export during low-magnitude events (Moatar et al., 2017).

Overall, our results indicate that water releases from Lake Okeechobee could be scheduled to potentially mitigate downstream nitrogen loading, but that modified scheduling may have little impact on downstream phosphorus concentrations. In particular, releases could be scheduled when dilution of  $NO_x$ , rather than mobilization, is more likely to happen (i.e., September–November), or when the response of  $NH_4^+$  is likely to be invariant as opposed to strongly positive (i.e., January, February, May through December). Unfortunately, these time periods do not overlap, demonstrating the complexity of establishing a schedule that mitigates all forms of downstream nutrient loading. The largely chemostatic behavior of TP and Ortho-P relative to discharge indicates that the Lake Okeechobee operation schedule may not be capable of being revised to appreciably impact downstream phosphorus concentrations in the CRE. This is in part related to the large legacy load of phosphorus in the lake (Reddy et al., 2011), which is reflected in the results of nutrient bioassay experiments, which show that nitrogen is the most commonly limiting nutrient for phytoplankton growth rates in the lake (Aldridge et al., 1995).

The relatively greater ability to influence nitrogen, as opposed to phosphorus, concentrations as a function of Lake Okeechobee operation scheduling could have nuanced repercussions for mitigating downstream water quality impacts, particularly harmful algal blooms (HABs). Previous studies report that the CRE can transition between nitrogen and phosphorus limitation in the freshwater portion of the system, while nitrogen limitation predominates in the saline regions of the CRE and nearshore Gulf of Mexico (Doering et al., 2006; Phlips et al., 2023). More specifically, nutrient loading to the CRE can affect HABs by: (a) driving the formation of autochthonous blooms within the estuary, and (b) enhancing nearshore red tides of toxic dinoflagellates, specifically through nitrogen loading (Medina et al., 2020, 2022; Phlips et al., 2023). In the latter scenario, the dominant red tide species along the southwest FL coast is the toxic dinoflagellate Karenia brevis (Heil et al., 2014a; Steidinger, 2009; Vargo, 2009). K. brevis is mixotrophic, and can take advantage of a wide range of nitrogen sources, including soluble inorganic and organic forms of nitrogen (Heil et al., 2014b), as well as particulate organic nitrogen through the direct phagotrophic consumption of pico/nanoplankton (Glibert et al., 2009). Several studies have observed the positive effect between nitrogen-levels in the CRE and the intensity of red tides (Medina et al., 2020, 2022; Phlips et al., 2023). Therefore, our findings indicate that there is some promise to adjust the Lake Okeechobee operation schedule to mitigate K. brevis blooms given the relationship between K. brevis concentrations in offshore waters and N concentrations from the CRE. However, autochthonous blooms occurring within the CRE, as opposed to in the nearshore environment, may require management of both N and P, depending on specific loading scenarios. The potential for algal blooms in the CRE can also be influenced by water residence time. During periods of sustained high discharge rates into the CRE, water residence times can be too short to support the accumulation of high biomass (Phlips et al., 2023). Therefore, the potential for autochthonous blooms in the CRE can be highest during low to moderate discharge rates.

When considering management actions that go beyond modifying operation schedule, our study's findings highlight how management efforts pursued in the surrounding C-43 basin could have great impact in terms of improving the quality of estuarine receiving waters. Large efforts are currently being undertaken to reduce nutrient loads from the C-43 basin to the CRE. Notably, several Basin Management Action Plans have been developed to implement Total Maximum Daily Loads (TMDLs), and the construction of the C-43 reservoir is currently underway (Caloosahatchee Estuary Basin Technical Stakeholders, 2012; Florida Department of Environmental Protection, 2020, 2022; Taylor et al., 2023). The C-43 reservoir aims to store excess water from the wet season when the CRE has too much freshwater and augment flow during the dry season. Previous C-43 reservoir modeling efforts have demonstrated that high flow releases could be reduced by 80% and nutrient loads by 30% (NRC, 2010). With the new construction of the C-43 reservoir, particular attention should be given to the

coordination of the C-43 reservoir and Lake Okeechobee operation schedules so as to not compromise water quality conditions of the downstream estuary. Additionally, future changes to the Lake Okeechobee operation schedule (i.e., LOSOM; Table 1) will allow for further exploration as to how modified release scheduling affects downstream nutrient concentrations, and the upcoming adoption of LOSOM also creates opportunities to co-ordinate its schedule with that of the C-43 reservoir.

Lastly, though we have focused on the role of C-43 watershed runoff and Lake Okeechobee releases on nutrient concentrations in the CRE, it is important to acknowledge that nutrient availability in the CRE is also influenced by inputs from the watershed directly adjacent to the estuary, also referred to as the "tidal basin" (Brewton et al., 2022; Rumbold & Doering, 2020). The relative balance of inputs from the tidal basin, upstream watershed, and Lake Okeechobee defines the nutrient signature of the CRE. In order to mitigate the effects of nutrient loading in the estuary, it is critical to understand all nutrient sources and their role in driving HABs and other water quality hazards. Furthermore, since the CRE is subject to both tidal mixing and strong river influences, the potential for autochthonous algal blooms is not only related to nutrient availability, but also to hydrologic conditions, such a water residence time (Mathews et al., 2015; Phlips et al., 2023; Sun et al., 2022; Wan et al., 2013). Thus, an understanding of the ways in which Lake Okeechobee water releases affect CRE residence time could also be useful for informing operation scheduling improvements to mitigate downstream water quality impacts. Finally, blooms in the CRE (i.e., allochthonous vs. autochthonous) (Reynolds et al., 2023), and understanding the risk of bloom export from Lake Okeechobee will require additional assessment of spatial and temporal dynamics of blooms within the lake itself (Tarabih et al., 2023).

## 5. Conclusions

To implement lake and reservoir management strategies that preserve downstream ecosystems, managers need to take into account the many nutrient sources to a downstream waterway, while also considering that nutrient contributions from different sources are affected by hydrologic conditions. This study investigated freshwater flow partitioning to the Caloosahatchee River and Estuary from a large managed lake, Lake Okeechobee, and the watershed associated with the river and estuary (i.e., the C-43 basin). The results demonstrate that Lake Okeechobee plays a significant role in defining nutrient concentrations in the CRE, but inputs from the C-43 basin watershed have relatively greater impacts on total water volume and nutrient concentrations. These findings highlight the importance of considering both lake and watershed management in efforts to improve water quality and reduce harmful algal blooms in downstream waters.

Our C-Q analyses revealed that NO<sub>x</sub> and NH<sub>4</sub><sup>+</sup> concentrations in the CRE were responsive to changes in discharge from Lake Okeechobee, while TP and Ortho-P concentrations were invariant to changes in discharge. Thus, modifications to the Lake Okeechobee operation schedule would predominantly affect downstream inorganic nitrogen concentrations in the CRE. Our study does not address whether reductions in nitrogen associated with modified discharge scheduling would significantly improve ecological conditions in the estuary, such as less intensive cyanobacteria and *K. brevis* blooms. Additionally, only concentrations, and not loads, were considered in this analysis, therefore some of the conclusions presented here may differ when analyzing loads. Moreover, we only considered nutrient concentrations measured from surface samples (depth  $\leq 0.5$  ft), and additional insights could be gained by analyzing measurements from samples collected at a range of depths to explore how nutrient concentrations, algal growth, dissolved oxygen concentrations, and the presence of submerged aquatic vegetation.

Overall, we conclude that modified lake release schedules have the potential to affect downstream nutrient concentrations, but the potential is limited by complex C-Q relationships and confounding effects from surrounding watersheds. The approaches and methods used in our study, such as C-Q analysis and the assessment of lake and watershed nutrient sources, could be applied to other freshwater systems confronting similar challenges with nutrient inputs from multiple sources, provided that sufficient data are available. Expanding this research to multiple sites could offer valuable insights into the variability of findings related to engineered water releases, phosphorus and nitrogen levels, and their impact on downstream waters. This broader investigation has the potential to enhance the generalizability of the study's findings across various environmental contexts, thereby contributing to the development of informed management strategies for the preservation of downstream ecosystems.

## **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

## **Data Availability Statement**

All data were accessed from DBHYDRO, a publicly available database maintained by the South Florida Water Management District (https://my.sfwmd.gov/dbhydroplsql/). Daily average discharge data were accessed for the S-77 and S-79 structures. Water quality parameters and daily total rainfall were also downloaded at the S-79 site. Analyzed data and a R script including analysis code are included in supplementary material.

#### References

- Aldridge, F., Phlips, E. J., & Schelske, C. L. (1995). The use of nutrient enrichment bioassays to test for spatial and temporal distribution of limiting factors affecting phytoplankton dynamics in Lake Okeechobee, Florida. Archive Hydrobiology, Advances in Limnology, 45, 177–190. Anderson, S. P., Dietrich, W. E., Torres, R., Montgomery, D. R., & Loague, K. (1997). Concentration-discharge relationships in runoff from a steep, unchanneled catchment. Water Resources Research, 33(1), 211–225. https://doi.org/10.1029/96wr02715
- Basu, N. B., Destouni, G., Jawitz, J. W., Thompson, S. E., Loukinova, N. V., Darracq, A., et al. (2010). Nutrient loads exported from managed catchments reveal emergent biogeochemical stationarity. *Geophysical Research Letters*, 37(23), L23404. https://doi.org/10.1029/ 2010gl045168
- Benson, N. G. (1981). The freshwater-inflow-to-estuaries issue. *Fisheries*, 6(5), 8–10. https://doi.org/10.1577/1548-8446(1981)006<0008: TFI>2.0.CO;2
- Birgand, F., Skaggs, R. W., Chescheir, G. M., & Gilliam, J. W. (2007). Nitrogen removal in streams of agricultural catchments—A literature review. Critical Reviews in Environmental Science and Technology, 37(5), 381–487. https://doi.org/10.1080/10643380600966426

Brewton, R. A., Kreiger, L. B., Tyre, K. N., Baladi, D., Wilking, L. E., Herren, L. W., & Lapointe, B. E. (2022). Septic system–groundwater– surface water couplings in waterfront communities contribute to harmful algal blooms in Southwest Florida. Science of the Total Environment, 837, 155319. https://doi.org/10.1016/j.scitotenv.2022.155319

Cadavid, L. G., Neidrauer, C. J., Obeysekera, J. T. B., Santee, E. R., Trimble, P., & Wilcox, W. (2012). Lake Okeechobee operations by means of the water supply and environment (WSE) regulation schedule. 166–175. https://doi.org/10.1061/40875(212)17

Caloosahatchee Estuary Basin Technical Stakeholders. (2012). Final basin management action plan for the implementation of total maximum daily loads for nutrients adopted by the Florida department of environmental protection in the Caloosahatchee Estuary Basin.

Cerco, C. F., & Noel, M. R. (2016). Impact of reservoir sediment scour on water quality in a downstream estuary. *Journal of Environmental Quality*, 45(3), 894–905. https://doi.org/10.2134/jeq2014.10.0425

- Charette, M. A., Henderson, P. B., Breier, C. F., & Liu, Q. (2013). Submarine groundwater discharge in a river-dominated Florida estuary. *Marine Chemistry*, 156, 3–17. https://doi.org/10.1016/j.marchem.2013.04.001
- Chen, J., Shi, H., Sivakumar, B., & Peart, M. R. (2016). Population, water, food, energy and dams. *Renewable and Sustainable Energy Reviews*, 56, 18–28. https://doi.org/10.1016/j.rser.2015.11.043

Dahm, C. N. (2010). Consequences of climate variability and human water demand on freshwater ecosystems: A Mediterranean perspective from the United States. *Water Scarcity in the Mediterranean: Perspectives Under Global Change*, 55–71.

Dewitz, J. (2021). National land cover database (NLCD) 2019 products [Dataset]. U.S. Geological Survey. https://doi.org/10.5066/P9KZCM54 Doering, P. H., & Chamberlain, R. H. (1999). Water quality and source of freshwater discharge to the Caloosahatchee Estuary, Florida. JAWRA Journal of the American Water Resources Association, 35(4), 793–806. https://doi.org/10.1111/j.1752-1688.1999.tb04175.x

Doering, P. H., Chamberlain, R. H., & Haunert, D. E. (2002). Using submerged aquatic vegetation to establish minimum and maximum freshwater inflows to the Caloosahatchee Estuary, Florida. *Estuaries*, 25(6), 1343–1354. https://doi.org/10.1007/bf02692229

Doering, P. H., Chamberlain, R. H., & Haunert, K. M. (2006). Chlorophyll a and its use as an indicator of eutrophication in the Caloosahatchee Estuary, Florida. *Florida Scientist*, 51–72.

- Douglass, J. G., Chamberlain, R. H., Wan, Y., & Doering, P. H. (2020). Submerged vegetation responses to climate variation and altered hydrology in a subtropical estuary: Interpreting 33 years of change. *Estuaries and Coasts*, 43(6), 1406–1424. https://doi.org/10.1007/s12237-020-00721-4
- Downing, J. A., Prairie, Y. T., Cole, J. J., Duarte, C. M., Tranvik, L. J., Striegl, R. G., et al. (2006). The global abundance and size distribution of lakes, ponds, and impoundments. *Limnology & Oceanography*, 51(5), 2388–2397. https://doi.org/10.4319/lo.2006.51.5.2388
- Duncan, J. M., Band, L. E., & Groffman, P. M. (2017). Variable nitrate concentration–discharge relationships in a forested watershed. Hydrological Processes, 31(9), 1817–1824. https://doi.org/10.1002/hyp.11136
- Fazekas, H. M., Wymore, A. S., & McDowell, W. H. (2020). Dissolved organic carbon and nitrate concentration-discharge behavior across scales: Land use, excursions, and misclassification. Water Resources Research, 56(8), e2019WR027028. https://doi.org/10.1029/2019WR027028
- Florida Department of Environmental Protection. (2020). Caloosahatchee River and Estuary Basin Management action plan. Retrieved from https://publicfiles.dep.state.fl.us/DEAR/DEARweb/BMAP/NEEP\_2020\_Updates/Caloosahatchee%20BMAP\_01-31-2020.pdf
- Florida Department of Environmental Protection. (2022). 2022 5-year review of the Caloosahatchee River and Estuary Basin management action plan. Retrieved from https://floridadep.gov/sites/default/files/Caloosahatchee%20BMAP%202022%205-Year%20Review\_.pdf
- Glibert, P. M., Burkholder, J. M., Kana, T. M., Alexander, J., Skelton, H., & Shilling, C. (2009). Grazing by *Karenia brevis* on Synechococcus enhances its growth rate and may help to sustain blooms. *Aquatic Microbial Ecology*, 55(1), 17–30. https://doi.org/10.3354/ame01279
- Godsey, S. E., Kirchner, J. W., & Clow, D. W. (2009). Concentration–discharge relationships reflect chemostatic characteristics of US catchments. *Hydrological Processes: International Journal*, 23(13), 1844–1864. https://doi.org/10.1002/hyp.7315
- Grill, G., Lehner, B., Thieme, M., Geenen, B., Tickner, D., Antonelli, F., et al. (2019). Mapping the world's free-flowing rivers. *Nature*, 569(7755), 215–221. https://doi.org/10.1038/s41586-019-1111-9
- Heil, C. A., Bronk, D. A., Dixon, L. K., Hitchcock, G. L., Kirkpatrick, G. J., Mulholland, M. R., et al. (2014a). The Gulf of Mexico ECOHAB: Karenia program 2006–2012. *Harmful Algae*, 38, 3–7. https://doi.org/10.1016/j.hal.2014.07.015

#### Acknowledgments

This work is supported by the U.S. Army Engineer Research and Development Center Cooperative Agreement W912HZ-21-2-0057.

- Heil, C. A., Dixon, L. K., Hall, E., Garrett, M., Lenes, J. M., O'Neil, J. M., et al. (2014b). Blooms of Karenia brevis (Davis) G. Hansen & Ø. Moestrup on the West Florida Shelf: Nutrient sources and potential management strategies based on a multi-year regional study. Harmful Algae, 38, 127–140. https://doi.org/10.1016/j.hal.2014.07.016
- Herndon, E. M., Dere, A. L., Sullivan, P. L., Norris, D., Reynolds, B., & Brantley, S. L. (2015). Landscape heterogeneity drives contrasting concentration–discharge relationships in shale headwater catchments. *Hydrology and Earth System Sciences*, 19(8), 3333–3347. https://doi. org/10.5194/hess-19-3333-2015
- Holland, R. A., Scott, K. A., Flörke, M., Brown, G., Ewers, R. M., Farmer, E., et al. (2015). Global impacts of energy demand on the freshwater resources of nations. *Proceedings of the National Academy of Sciences of the United States of America*, 112(48), E6707–E6716. https://doi.org/ 10.1073/pnas.1507701112
- Huang, J., Zhan, J., Yan, H., Wu, F., & Deng, X. (2013). Evaluation of the impacts of land use on water quality: A case study in the Chaohu Lake Basin. *The Scientific World Journal*, 2013, e329187. https://doi.org/10.1155/2013/329187
- Jones, C. S., Schilling, K. E., Simpson, I. M., & Wolter, C. F. (2018). Iowa stream nitrate, discharge and precipitation: 30-year perspective. *Environmental Management*, 62(4), 709–720. https://doi.org/10.1007/s00267-018-1074-x
- Julian, P., & Osborne, T. Z. (2018). From lake to estuary, the tale of two waters: A study of aquatic continuum biogeochemistry. Environmental Monitoring and Assessment, 190(2), 96. https://doi.org/10.1007/s10661-017-6455-8
- Julian, P., & Reidenbach, L. (2023). Upstream water management and its role in estuary health, evaluation of freshwater management and subtropical estuary function. (Preprint). In Review. https://doi.org/10.21203/rs.3.rs-2565249/v1

Kendall, M. G. (1948). Rank correlation methods.

- Klatt, J. G., Mallarino, A. P., Downing, J. A., Kopaska, J. A., & Wittry, D. J. (2003). Soil phosphorus, management practices, and their relationship to phosphorus delivery in the Iowa Clear Lake agricultural watershed. *Journal of Environmental Quality*, 32(6), 2140–2149. https://doi.org/10. 2134/jeq2003.2140
- Knapp, J. L. A., von Freyberg, J., Studer, B., Kiewiet, L., & Kirchner, J. W. (2020). Concentration–discharge relationships vary among hydrological events, reflecting differences in event characteristics. *Hydrology and Earth System Sciences*, 24(5), 2561–2576. https://doi.org/10. 5194/hess-24-2561-2020
- Kramer, B. J., Davis, T. W., Meyer, K. A., Rosen, B. H., Goleski, J. A., Dick, G. J., et al. (2018). Nitrogen limitation, toxin synthesis potential, and toxicity of cyanobacterial populations in Lake Okeechobee and the St. Lucie River Estuary, Florida, during the 2016 state of emergency event. *PLoS One*, 13(5), e0196278. https://doi.org/10.1371/journal.pone.0196278
- Lehner, B., Liermann, C. R., Revenga, C., Vörösmarty, C., Fekete, B., Crouzet, P., et al. (2011). High-resolution mapping of the world's reservoirs and dams for sustainable river-flow management. Frontiers in Ecology and the Environment, 9(9), 494–502. https://doi.org/10.1890/100125
- Ling, T.-Y., Gerunsin, N., Soo, C.-L., Nyanti, L., Sim, S.-F., & Grinang, J. (2017). Seasonal changes and spatial variation in water quality of a large young tropical reservoir and its downstream river. *Journal of Chemistry*, 2017, e8153246. https://doi.org/10.1155/2017/8153246
- Liu, W., Tian, S., Youssef, M. A., Birgand, F. P., & Chescheir, G. M. (2022). Patterns of long-term variations of nitrate concentration–Stream discharge relationships for a drained agricultural watershed in Mid-western USA. *Journal of Hydrology*, 614, 128479. https://doi.org/10. 1016/j.jhydrol.2022.128479
- Liu, Z., Choudhury, S. H., Xia, M., Holt, J., Wallen, C. M., Yuk, S., & Sanborn, S. C. (2009). Water quality assessment of coastal Caloosahatchee River watershed, Florida. Journal of Environmental Science and Health, Part A, 44(10), 972–984. https://doi.org/10.1080/ 10934520902996872
- Mann, H. B. (1945). Nonparametric tests against trend. Econometrica: Journal of the Econometric Society, 13(3), 245–259. https://doi.org/10. 2307/1907187
- Mathews, A. L., Phlips, E. J., & Badylak, S. (2015). Modeling phytoplankton productivity in a shallow, microtidal, subtropical estuary. Marine Ecology Progress Series, 531, 63–80. https://doi.org/10.3354/meps11313
- Medina, M., Huffaker, R., Jawitz, J. W., & Muñoz-Carpena, R. (2020). Seasonal dynamics of terrestrially sourced nitrogen influenced Karenia brevis blooms off Florida's southern Gulf Coast. Harmful Algae, 98, 101900. https://doi.org/10.1016/j.hal.2020.101900
- Medina, M., Kaplan, D., Milbrandt, E. C., Tomasko, D., Huffaker, R., & Angelini, C. (2022). Nitrogen-enriched discharges from a highly managed watershed intensify red tide (*Karenia brevis*) blooms in southwest Florida. Science of the Total Environment, 827, 154149. https://doi. org/10.1016/j.scitotenv.2022.154149
- Minaudo, C., Dupas, R., Gascuel-Odoux, C., Roubeix, V., Danis, P.-A., & Moatar, F. (2019). Seasonal and event-based concentration-discharge relationships to identify catchment controls on nutrient export regimes. Advances in Water Resources, 131, 103379. https://doi.org/10.1016/j. advwatres.2019.103379
- Moatar, F., Abbott, B. W., Minaudo, C., Curie, F., & Pinay, G. (2017). Elemental properties, hydrology, and biology interact to shape concentration-discharge curves for carbon, nutrients, sediment, and major ions. *Water Resources Research*, 53(2), 1270–1287. https://doi.org/ 10.1002/2016WR019635
- Motew, M., Chen, X., Booth, E. G., Carpenter, S. R., Pinkas, P., Zipper, S. C., et al. (2017). The influence of legacy P on lake water quality in a Midwestern agricultural watershed. *Ecosystems*, 20(8), 1468–1482. https://doi.org/10.1007/s10021-017-0125-0
- Musolff, A., Zhan, Q., Dupas, R., Minaudo, C., Fleckenstein, J. H., Rode, M., et al. (2021). Spatial and temporal variability in concentrationdischarge relationships at the event scale. Water Resources Research, 57(10), e2020WR029442. https://doi.org/10.1029/2020wr029442
- Nilsson, C., Reidy, C. A., Dynesius, M., & Revenga, C. (2005). Fragmentation and flow regulation of the world's large river systems. *Science*, 308(5720), 405–408. https://doi.org/10.1126/science.1107887
- NRC. (2000). Clean coastal waters: Understanding and reducing the effects of nutrient pollution. National Academies Press.
- NRC. (2010). Progress toward restoring the Everglades: The 3rd biennial review. The National Academies Press.
- Paerl, H. W., Joyner, J. J., Joyner, A. R., Arthur, K., Paul, V., O'Neil, J. M., & Heil, C. A. (2008). Co-occurrence of dinoflagellate and cyanobacterial harmful algal blooms in southwest Florida coastal waters: Dual nutrient (N and P) input controls. *Marine Ecology Progress Series*, 371, 143–153. https://doi.org/10.3354/meps07681
- Peterson, B. J., Wollheim, W. M., Mulholland, P. J., Webster, J. R., Meyer, J. L., Tank, J. L., et al. (2001). Control of nitrogen export from watersheds by headwater streams. *Science*, 292(5514), 86–90. https://doi.org/10.1126/science.1056874
- Phlips, E. J., Badylak, S., Hart, J., Haunert, D., Lockwood, J., Manley, H., et al. (2012). Climatic influences on autochthonous and allochthonous phytoplankton blooms in a subtropical estuary, St. Lucie Estuary, Florida, USA. *Estuaries and Coasts*, 35(1), 335–352. https://doi.org/10.1007/ s12237-011-9442-2
- Phlips, E. J., Badylak, S., Mathews, A. L., Milbrandt, E. C., Montefiore, L. R., Morrison, E. S., et al. (2023). Algal blooms in a river-dominated estuary and nearshore region of Florida, USA: The influence of regulated discharges from water control structures on hydrologic and nutrient conditions. *Hydrobiologia*, 850(20), 4385–4411. https://doi.org/10.1007/s10750-022-05135-w

- Phlips, E. J., Badylak, S., Nelson, N. G., & Havens, K. E. (2020). Hurricanes, El Niño and harmful algal blooms in two sub-tropical Florida estuaries: Direct and indirect impacts. *Scientific Reports*, 10(1), 1910. https://doi.org/10.1038/s41598-020-58771-4
- Poff, N. L., Olden, J. D., Merritt, D. M., & Pepin, D. M. (2007). Homogenization of regional river dynamics by dams and global biodiversity implications. Proceedings of the National Academy of Sciences of the United States of America, 104(14), 5732–5737. https://doi.org/10.1073/ pnas.0609812104
- R Core Team. (2020). A language and environment for statistical computing (3.6.3). Retrieved from https://www.R-project.org/
- Rechcigl, J. E. (1997). Phosphorus—Natural versus pollution levels: Lake Okeechobee case study. In 46th Proceedings of Annual Florida Beef Cattle Short Course (p. 61).
- Reddy, K. R., Newman, S., Osborne, T. Z., White, J. R., & Fitz, H. C. (2011). Phosphorous cycling in the greater Everglades ecosystem: Legacy phosphorous implications for management and restoration. *Critical Reviews in Environmental Science and Technology*, 41(S1), 149–186. https://doi.org/10.1080/10643389.2010.530932
- Reynolds, N., Schaeffer, B. A., Guertault, L., & Nelson, N. G. (2023). Satellite and in situ cyanobacteria monitoring: Understanding the impact of monitoring frequency on management decisions. *Journal of Hydrology*, 619, 129278. https://doi.org/10.1016/j.jhydrol.2023.129278
- Rose, L. A., Karwan, D. L., & Godsey, S. E. (2018). Concentration–discharge relationships describe solute and sediment mobilization, reaction, and transport at event and longer timescales. *Hydrological Processes*, 32(18), 2829–2844. https://doi.org/10.1002/hyp.13235
- Rumbold, D. G., & Doering, P. H. (2020). Water quality and source of freshwater discharge to the Caloosahatchee Estuary, Florida: 2009–2018. *Florida Scientist*, 83(1), 1–20. https://www.jstor.org/stable/26975620
- Scott, T. M., & Missimer, T. M. (2001). Geology and hydrology of Lee County, Florida. Florida Geological Survey Special Publication. Retrieved from https://www.researchgate.net/publication/280112317\_The\_surficial\_geology\_of\_Lee\_County\_and\_the\_Caloosahatchee\_Basin
- SFWMD. (2009). Caloosahatchee River watershed protection plan (p. 276). South Florida Water Management District, Florida Department of Environmental Protection, Florida Department of Agriculture and Consumer Services. Retrieved from https://www.sfwmd.gov/sites/default/files/documents/ne\_crwpp\_main\_123108.pdf
- Sklar, F. H., Chimney, M. J., Newman, S., McCormick, P., Gawlik, D., Miao, S., et al. (2005). The ecological–societal underpinnings of Everglades restoration. *Frontiers in Ecology and the Environment*, 3(3), 161–169. https://doi.org/10.1890/1540-9295(2005)003[0161:teuoer]2.0. co;2
- Steidinger, K. A. (2009). Historical perspective on Karenia brevis red tide research in the Gulf of Mexico. Harmful Algae, 8(4), 549–561. https://doi.org/10.1016/j.hal.2008.11.009
- Sun, D., Barton, M., Parker, M., & Sheng, Y. P. (2022). Estuarine water quality: One-dimensional model theory and its application to a riverine subtropical estuary in Florida. *Estuarine, Coastal and Shelf Science*, 277, 108058. https://doi.org/10.1016/j.ecss.2022.108058
- Tarabih, O. M., & Arias, M. E. (2021). Hydrological and water quality trends through the lens of historical operation schedules in Lake Okeechobee. Journal of Water Resources Planning and Management, 147(7), 04021034. https://doi.org/10.1061/(asce)wr.1943-5452.0001395
- Tarabih, O. M., Dang, T. D., Paudel, R., & Arias, M. E. (2023). Lake operation optimization of nutrient exports: Application of phosphorus control in the largest subtropical lake in the United States. *Environmental Modelling & Software*, 160, 105603. https://doi.org/10.1016/j.envsoft.2022. 105603
- Taylor, D., Parker, M., Armstrong, C., Bobsein, J., & Barton, M. (2023). Caloosahatchee River watershed protection plan annual progress report. USACE. (2022). National inventory of dams. Retrieved from https://nid.sec.usace.army.mil/#/
- Vargo, G. A. (2009). A brief summary of the physiology and ecology of Karenia brevis Davis (G. Hansen and Moestrup comb. Nov.) red tides on the West Florida Shelf and of hypotheses posed for their initiation, growth, maintenance, and termination. Harmful Algae, 8(4), 573–584. https://doi.org/10.1016/j.hal.2008.11.002
- Volety, A. K., Savarese, M., Tolley, S. G., Arnold, W. S., Sime, P., Goodman, P., et al. (2009). Eastern oysters (Crassostrea virginica) as an indicator for restoration of Everglades ecosystems. *Ecological Indicators*, 9(6), S120–S136. https://doi.org/10.1016/j.ecolind.2008.06.005
- Vörösmarty, C. J., McIntyre, P. B., Gessner, M. O., Dudgeon, D., Prusevich, A., Green, P., et al. (2010). Global threats to human water security and river biodiversity. *Nature*, 467(7315), 555–561. Article 7315. https://doi.org/10.1038/nature09440
- Wan, Y., Qiu, C., Doering, P., Ashton, M., Sun, D., & Coley, T. (2013). Modeling residence time with a three-dimensional hydrodynamic model: Linkage with chlorophyll a in a subtropical estuary. *Ecological Modelling*, 268, 93–102. https://doi.org/10.1016/j.ecolmodel.2013.08.008
- World Commission on Dams (WCD). (2000). Dams and development: A new framework for decision-making: The report of the world commission on dams. Earthscan.
- Wu, H., Chen, J., Xu, J., Zeng, G., Sang, L., Liu, Q., et al. (2019). Effects of dam construction on biodiversity: A review. Journal of Cleaner Production, 221, 480–489. https://doi.org/10.1016/j.jclepro.2019.03.001
- Yi, Y., Yang, Z., & Zhang, S. (2010). Ecological influence of dam construction and river-lake connectivity on migration fish habitat in the Yangtze River basin, China. Procedia Environmental Sciences, 2, 1942–1954. https://doi.org/10.1016/j.proenv.2010.10.207
- Zacharias, Q., & Kaplan, D. (2023). The past, present, and potential future of phosphorus management in the Florida Everglades. *Restoration Ecology*, 31(3), e13799. https://doi.org/10.1111/rec.13799